Author's response to reviews

Title: Cardiac calcium score on 2D echo: correlations with cardiac and coronary calcium at multi-detector computed tomography.

Authors:

Nicola Gaibazzi (ngaibazzi@gmail.com)
Chiara Baldari (chiara.baldari@email.it)
Pompilio Faggiano (faggiano@numerica.it)
Lisa Albertini (lisa.albertini@alice.it)
Giacomo Faden (gfaden@hotmail.com)
Filippo Pigazzani (filippo.pigazzani@gmail.com)
Cristina Rossi (crossi@unipr.it)
Claudio Reverberi (CReverberi@ao.pr.it)

Version: 2 Date: 6 October 2014

Author's response to reviews:

Rosa Sicari, MD October 6, 2014
Editor Cardiovascular Ultrasound

Re – Cardiac calcium score on 2D echo: correlations with cardiac and coronary calcium at multi-detector computed tomography.

Dear Prof Sicari,

We would like to submit the above - We are attaching the point by point rebuttal letter.

Thanks

Nicola Gaibazzi